Title: EFFECTIVENESS OF PHYSICAL EXERCISE IN THE TREATMENT OF DEPRESSION IN OLDER ADULTS AS AN ALTERNATIVE TO ANTIDEPRESSANT DRUGS IN PRIMARY CARE

Authors:

Jesús López-Torres Hidalgo (jesusl@sescam.org)
Luis Aguilar Salmerón (geniusluis10@gmail.com)
Clotilde Boix Gras (cboix@sescam.org)
Monchi Campos Rosa (monchi.campos@gmail.com)
Francisco Escobar Rabadán (fescobarr@yahoo.es)
Concepción Escolano Vizcaíno (concepcion@sescam.jccm.es)
José Luis Estellés Belenguer (jlestelles@sescam.jccm.es)
Juan Fernández Martín (jfernandezm@jccm.es)
Vicente Ferrer López (ferrerlopezv@gmail.com)
Almudena Legido García (alegido@sescam.jccm.es)
Jaime López-Torres López (jaime.lopeztorres.lopez@gmail.com)
Jesús López-Torres López (jlopeztl@yahoo.es)
María Ángeles López Verdejo (malopezv@sescam.jccm.es)
Ana López Yeste (anloye@sescam.jccm.es)
María Ángeles Lloret Callejo (malloret@sescam.org)
María Jesús Montes Lozano (mjmlxi@gmail.com)
Juana Muñoz Núñez (jmunozn@sescam.jccm.es)
Karen Nieto Rodríguez (Knieto@yahoo.com)
Joseba Rabanales Sotos (Joseba.Rabanales@uclm.es)
I send you again the manuscript "EFFECTIVENESS OF PHYSICAL EXERCISE IN THE TREATMENT OF DEPRESSION IN OLDER ADULTS AS AN ALTERNATIVE TO ANTIDEPRESSANT DRUGS IN PRIMARY CARE" (BPSY-D-18-00264).

To the email address BMCSeriesEditorial@biomedcentral.com I have sent an approval document of the Clinical Research Ethics Committee, documents on the granting of funding by the Carlos III Health Institute of Madrid (Resolution granting the financial assistance) and certificate of the Coordinator of the Center where the study is being carried out, Dr. José Luis Estellés.

This is a study that received external funding from the Carlos III Health Institute in the call for grants for Health Research Projects of 2015 (file number PI15 / 01486). These aids are obtained through competitive channels and through a peer review process.

A SPIRIT checklist has been sent.

The study is currently underway and the researchers are finalizing the data collection. In the last quarter of 2018 we hope to have completed this phase of the study and we will be able to obtain the first results.

All the authors fulfilled the necessary criteria to be considered as such and we all agree to be part of the authorship of the manuscript, as well as the order of the authors.

Best Regards

Jesús López-Torres